

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Hope, Ole-Kristian

#### **Article**

Large shareholders and accounting research

China Journal of Accounting Research

# **Provided in Cooperation with:**

Sun Yat-sen University

Suggested Citation: Hope, Ole-Kristian (2013): Large shareholders and accounting research, China Journal of Accounting Research, ISSN 1755-3091, Elsevier, Amsterdam, Vol. 6, Iss. 1, pp. 3-20, https://doi.org/10.1016/j.cjar.2012.12.002

This Version is available at: https://hdl.handle.net/10419/187583

# Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/3.0/





Contents lists available at SciVerse ScienceDirect

# China Journal of Accounting Research

journal homepage: www.elsevier.com/locate/cjar



# Large shareholders and accounting research<sup>☆</sup>

# Ole-Kristian Hope

University of Toronto, Rotman School of Management, Canada

ARTICLE INFO

Article history:
Received 5 April 2012
Accepted 11 July 2012
Available online 29 January 2013

JEL classification:

G30

G32

G38

M20 M41

Keywords: Large shareholders Agency costs Controlling owners

Minority owners
Private firms

International

ABSTRACT

Large shareholders are a potentially very important element of firms' corporate governance system. Whereas analytical research is typically vague on who these large shareholders are, in practice there are important variations in the types of large owners (and the different types of large owners could play very different governance roles). After briefly reviewing the standard agency cost arguments, in this article I emphasize the heterogeneity of concentrated ownership and in particular focus on the roles of families, institutions, governments, and employee ownership. I also discuss the role of large shareholders in private (i.e., unlisted) firms, where ownership tends to be more concentrated than in publicly traded firms. Finally, I briefly discuss variations in ownership structures across selected countries.

© 2013 China Journal of Accounting Research. Founded by Sun Yat-sen University and City University of Hong Kong. Production and hosting by Elsevier B.V. All rights reserved.

#### 1. Introduction

This article is based on my keynote address at the 2012 CJAR Special Issue Symposium at CEIBS in Shanghai. The topic of the conference was "large shareholders" and I was honored to be given the opportunity to

E-mail addresses: e.xia@elsevier.com, okhope@rotman.utoronto.ca



Production and hosting by Elsevier

 $1755-3091/\$-see front matter © 2013 \ China \ Journal of \ Accounting \ Research. \ Founded \ by \ Sun \ Yat-sen \ University \ and \ City \ University \ of \ Hong \ Kong. \ Production \ and \ hosting \ by \ Elsevier \ B.V. \ All \ rights \ reserved. \ http://dx.doi.org/10.1016/j.cjar.2012.12.002$ 

<sup>\*</sup> This article has been prepared for the *China Journal of Accounting Research* and forms the basis for my CJAR Special Issue Symposium presentation (March 30, 2012). I appreciate useful comments from Heather Li and acknowledge the financial support of the Deloitte Professorship.

make some comments on how large shareholders are important for (accounting) research. I should hasten to say that there are several well-cited survey studies on corporate governance in accounting, economics, finance, and management. Thus, in this paper I will not attempt a complete survey on the literature on large shareholders. Instead, I have decided to focus on one particular aspect – the *heterogeneity* of large shareholders.

We tell our PhD students that they should base their research on theory to the extent possible. At least in financial accounting the "theory" that is referred to is often analytical economics-based research. At the Rotman School we have the same emphasis on theory and I am personally a strong believer in anchoring your work in theory. However, most analytical models are vague (to put it mildly) when describing exactly who the large shareholders are and how they act. As this article will highlight, there is in fact rather considerable diversity in the *types of large shareholders* we observe, and it is very likely that these may have different effects on outcomes of interest to accounting researchers. Hence the reader can consider this article also as a call for "attention to the context" in which the study is conducted. For example, I would encourage "case-based" type studies that delve deeper into one particular form of large shareholder, such as state-owned enterprises in China.

I would like to offer three brief caveats. First, as already mentioned there are other, more comprehensive surveys on corporate governance issues and I would recommend that readers consult these if relevant. Second, although I consider several different types of large shareholders I could clearly have included additional types (e.g., the effect of foreign shareholders). Finally, there are important measurement issues in defining large shareholders (using cut-offs; multiple large owners; concentration ratios; ownership percentage versus voting rights; considering potential nonlinearities; organizational form; etc.).

Section 2 provides a brief review of the classic Jensen and Meckling (1976) arguments and discusses both vertical and horizontal agency costs. It also discusses the role of the second-largest shareholders and examines how large shareholders exercise their monitoring in practice. Section 3 focuses on who the large shareholders are. The chapter considers the roles of families, institutions, governments, and employee ownership. Large shareholders are particularly prominent in private (i.e., unlisted) firms, and Section 4 summarizes relevant research on these economically very important firms. Section 5 contains a discussion of variations across selected countries in the types of dominating ownership, and Section 6 concludes.

#### 2. Overview of large shareholders and agency costs

#### 2.1. Brief review of Jensen and Meckling (1976)

As this conference is motivated to a large extent by Jensen and Meckling (1976), it is worthwhile to first briefly revisit and review their seminal study. Jensen and Meckling define an agency relationship as a contract under which one or more persons (the principal(s)) engage another person (the agent) to perform some service on their behalf which involves delegating some decision making authority to the agent. If both parties are utility maximizers there is good reason to believe that the agent will not always act in the best interests of the principal. The principal can limit divergences from his interest by establishing appropriate incentives for the agent and by incurring monitoring costs designed to limit the value-reducing activities of the agent.<sup>2</sup>

If a wholly owned firm is managed by the owner, he will make decisions which maximize his utility. This situation is of course unusual other than for the smallest private firms and by definition not observed in publicly traded companies. In such cases, Jensen and Meckling argue that agency costs will be generated by the divergence between his interest and those of the outside shareholders, as he will then bear only a fraction of the costs of any non-pecuniary benefits he takes out in maximizing his own utility. Put differently, as the owner—manager's fraction of the equity falls, his fractional claim on the outcomes falls and this will tend to encourage him to appropriate larger amounts of the corporate resources in the form of perquisites. This also makes it

<sup>&</sup>lt;sup>1</sup> Jensen and Meckling's article was in part motivated by the observation by Adam Smith (1776) that "The directors of such [joint-stock] companies, however, being the managers rather of other people's money than of their own, it cannot be well expected, that they should watch over it with the same anxious vigilance with which the partners in a private copartnery frequently watch over their own... Negligence and profusion, therefore, must always prevail, more or less, in the management of the affairs of such a company."

<sup>&</sup>lt;sup>2</sup> In some situations it will pay the agent to expend resources to guarantee that he will not take certain actions which would harm the principal or to ensure that the principal will be compensated if he does take such actions (referred to as "bonding").

desirable for the minority shareholders to expend more resources in monitoring his behavior.<sup>3</sup> Important for us in accounting, this is clearly one of the reasons for the demand for accounting-related information. In fact, the genesis of accounting was in the "stewardship role" it can play in monitoring agents (see, e.g., Gjesdal, 1981 for a nice discussion). It is only more recently that the "valuation role" of accounting information has gained in prominence (and may well be the dominating role today). Related to the stewardship role (or governance) role of accounting, Jensen and Meckling argue that their theory can explain "why accounting reports would be provided voluntarily to creditors and stockholders, and why independent auditors would be engaged by management to testify to the accuracy and correctness of such reports."

# 2.2. More on the role of ownership concentration (or importance of large shareholders)

There are two common approaches to corporate governance throughout most of the world (e.g., Shleifer and Vishny, 1997). First, investors' rights are protected to varying degrees across the world through the legal process and legal environment. The second major approach, and the focus of this article, is ownership by large investors.

Research provides evidence that managers, when left unmonitored, are more likely to manage earnings, commit fraud, or make suboptimal investment decisions (e.g., Biddle and Hilary, 2006; Hope and Thomas, 2008). Thus, shareholder monitoring is an important mechanism by which agency costs can be reduced. However, while all shareholders have the responsibility to monitor managerial activities, the benefits of doing so by any individual shareholder are proportional to the percentage of shares owned (Jensen and Meckling, 1976; Shleifer and Vishny, 1997). Put another way, when ownership is widely dispersed, it is economically less feasible for any individual shareholder to incur significant monitoring costs, because she will receive only a small portion of benefits. Similarly, when ownership is dispersed, it is harder for shareholders to monitor managerial actions.

Thus, as the percentage of ownership by individual shareholders increases (i.e., concentration increases), the more willing individual shareholders are to incur necessary monitoring costs. That is, when ownership is limited to one or a few individuals, it is easier and more efficient for those individuals to directly monitor managerial actions. This is the typical "vertical agency cost" argument (i.e., conflicts between managers and owners) and leads to the general prediction that agency costs are expected to be lower as ownership concentration increases.<sup>4</sup>

Potential manager—owner conflicts are not the only relevant issues. Horizontal agency costs relate to how large shareholders can decrease a firm's value through extracting private benefit from the minority shareholders (e.g., La Porta et al., 1999). Morck et al. (1988) argue that increased ownership concentration may entrench managers, as they are increasingly less subject to governance by boards of directors and to discipline by the market for corporate control. Controlling shareholders may either engage in outright expropriation from self-dealing transactions or exercise de facto expropriation through the pursuit of objectives that are not profit-maximizing in return for personal utilities. These controlling shareholders may attempt to hide these activities from other stakeholders (e.g., minority shareholders and creditors) by manipulating reported performance (an issue of obvious interest to accountants). In other words, a controlling owner can increase agency costs via the positive association with private benefits of control (e.g., Hope et al., 2012a).

To summarize the discussion, the presence of a controlling owner represents forces that work in opposite directions. For a researcher, this is both a challenge and an opportunity. It is an opportunity if the researcher is able to specify ex ante which set of agency costs is likely to be most significant. For example, in countries with less legal protection of the minority shareholders the main agency problem often exists between controlling shareholders and minority shareholders.

<sup>&</sup>lt;sup>3</sup> Jensen and Meckling consider the term monitoring to include more than just measuring or observing the behavior of the agent. It includes efforts on the part of the principal to "control" the behavior of the agent through budget restrictions, compensation policies, operating rules, etc.

<sup>&</sup>lt;sup>4</sup> Furthermore, controlling shareholders could enable a long investment horizon which allows the building of strong relationships between the firms and outside providers of capital (Ellul et al., 2009). In fact, a controlling shareholder could increase business focus and make contracting negotiations easier.

#### 2.3. The role of the second-largest shareholder

While the previous discussion explains the need for shareholders to monitor managers, the literature also establishes the need for shareholders to monitor one another. For example, controlling shareholders have the ability to exploit minority shareholders in closely-held corporations (e.g., Nagar et al., 2011). Such exploitation can include higher compensation to controlling shareholders, misappropriation of assets, and dilution of minority shareholders' interests through the issuance of stock or dividends (Gogineni et al., 2010). As the ownership stake of a second shareholder increases, so does her ability and willingness to effectively monitor the largest shareholder. The monitoring activities by the second largest shareholder would be similar to those used by the largest shareholder to monitor managers (Hope et al., 2012a).

Pagano and Roell (1998) specify conditions under which large shareholders monitor each other, reducing expropriation and improving firm performance. They predict that expropriation of minority shareholders is likely to be less severe when the ownership stake of non-controlling shareholders is more concentrated, as such concentration makes it easier and more effective to monitor the controlling shareholder. This is the typical "horizontal agency cost" argument (i.e., conflicts between majority and minority shareholders) and leads to the prediction that as ownership by the second largest shareholder increases, agency costs decrease.

#### 2.4. How do large shareholders exercise their monitoring?

Often finance and accounting research is vague on the *mechanisms* through which monitoring happens. In practice monitoring by a large shareholder could take many forms. Perhaps the most commonly discussed means of monitoring discussed in the literature involves a large shareholder having a seat on the board. Several studies show in a variety of contexts the board's role in monitoring managers (e.g., Fama, 1980; Fama and Jensen, 1983; Adams et al., 2010). Other forms of direct monitoring would be a large shareholder actively participating in the firm's operations or having routine meetings with managers. As the proportion of ownership increases, the more beneficial it is for large shareholders to engage in these types of costly direct monitoring activities. Large shareholders can also serve to block business decisions that may be considered suboptimal (e.g., aggressive expansion through negative net present value projects). Doing so involves an investment in time and expertise by the shareholder to understand the consequences of major business decisions. Large shareholders are also likely to have more control over the firm's dividend (or capital distribution) policy, as a way to further discipline managers' actions.

#### 3. Who are the large shareholders? Does it matter?

Analytical research on large shareholders tends to be rather generic and often does not consider that there may be very different types of large shareholders. There is surprisingly limited extant research on how different groups of large shareholders can affect corporate outcomes (e.g., financial reporting quality).<sup>5</sup> Here I briefly consider research on the following owner types: families (including the CEO as owner), institutional investors, governments, and employees.

#### 3.1. Family ownership

A large fraction of businesses throughout the world are organized around families and there is a relatively large literature on family ownership (Bertrand and Schoar, 2006). Most of this research in on publicly listed companies. For example, family-controlled firms dominate in East Asia and Latin America. As an indication of the importance of family firms, La Porta et al. (1999) report that 65% of the 20 largest firms in Argentina have at least a 20% family stake; in Hong Kong this fraction is 70%. In contrast, in Japan the corresponding number is 5%.

<sup>&</sup>lt;sup>5</sup> An exception is Cronqvist and Fahlenbrach (2009). They examine effects of different types of institutional investors in the US and find that investor type has significant effects on several corporate policies. The only study I'm aware of in accounting is Dou et al. (2012) who follow an approach similar to Cronqvist and Fahlenbrach (2009) and examine the effects of large shareholders on accounting practices for a large sample of US firms over the 2001–2009 period.

If a researcher is really interested in examining the effects of family ownership, it would seem that private (i.e., unlisted) firms offer even more fertile ground for research. Section 4 discusses private firms in more detail.

A stream of research has examined "family firms" included in the S&P 500. This line of research is primarily motivated by the fact that, notwithstanding the oft-cited idea that US publicly firms have widely dispersed ownership, Shleifer and Vishny (1986) (and others) document that large shareholders are common and, in particular, note that founding families continue to hold equity stakes and board seats in nearly 33% of the Fortune 500 firms. In other words, US firms may not be as different from those observed elsewhere in the world as thought by many. These founding families represent a unique class of long-term shareholders that hold poorly diversified portfolios and often control senior management positions. Family owners can thus exert influence and control over the firm, potentially leading to performance differences with nonfamily firms.

In a widely cited study, Anderson and Reeb (2003) investigate the relation between founding-family ownership and firm performance. They find that, contrary to their conjecture, family firms perform better than nonfamily firms. Additional analyses reveal that the relation between family holdings and firm performance is nonlinear and that when family members serve as CEO, performance is better than with outside CEOs. Overall, their results are inconsistent with the hypothesis that minority shareholders are adversely affected by family ownership, suggesting that family ownership may be an effective organizational structure.

Ali et al. (2007) recognize that, compared with nonfamily firms, family firms face less severe agency problems due to the separation of ownership and management. However, they face more severe agency problems that arise between controlling and non-controlling shareholders. These conflicting effects are often referred to as "entrenchment versus alignment." Thus it is not clear what to predict regarding family firms' disclosure practices relative to other firms. Using a sample of only S&P 500 firms, Ali et al. (2007) conclude that family firms report better quality earnings, are more likely to warn for a given magnitude of bad news, but make fewer disclosures about their corporate governance practices. Consistent with family firms making better financial disclosures, the authors find that family firms have larger analyst following, more informative analysts' forecasts, and smaller bid–ask spreads.<sup>6</sup>

It is far from clear that the above findings should be generalized to other settings, even in the United States. First, although the firms classified as "family firms" by definition meet the definition of a family firm for these studies, others may employ a higher threshold for family ownership. Given the nonlinearities documented in the US setting, it is thus highly unclear what to expect in very different environments and with much higher family ownership (e.g., in private firms). Even more importantly, conflicting evidence exists on whether having family ownership increases or decreases a firm's value, and it seems to be country dependent. Bertrand and Schoar (2006) conclude that there is no strong empirical evidence for the economic superiority of family-controlled businesses. According to Bertrand and Schoar (2006), family firms appear to underperform relative to nonfamily firms in most countries: for example, Claessens et al. (2002) for several Southeast Asian countries; Morck et al. (2000) for Canada; and Cronqvist and Nilsson (2003) for Sweden. Also, Bloom and Van Reenen (2007) find that family firms in France, Germany, the United Kingdom and the United States are systematically associated with worse managerial practices. Bertrand and Schoar (2006) note two important exceptions. Khanna and Palepu (2000) find that business groups in India, which are for the most part family-controlled, perform better than stand-alone firms in matched industries (see more on this below); and Sraer and Thesmar (2007) who find a premium for family firms in France.

# 3.1.1. The role of the CEO in family firms

There is comparatively limited research on the role of the CEO as part of the dominant family. A dominant belief in the literature is that as CEO ownership increases, her incentives align more with those of other shareholders, reducing the agency problem that arises from separation of ownership and control (e.g., Jensen and Meckling, 1976). This is known as the alignment effect which suggests reduced agency costs.

<sup>&</sup>lt;sup>6</sup> In a closely related study, Chen et al. (2008) find that, compared with nonfamily firms, family firms provide fewer earnings forecasts and conference calls, but more earnings warnings. The authors interpret the former to be consistent with family owners having a longer investment horizon, better monitoring of management, and lower information asymmetry between owners and managers, they interpret the higher likelihood of earnings warnings to be consistent with family owners having greater litigation and reputation cost concerns. In another related paper, Wang (2006) finds that founding family ownership is associated with higher earnings quality in S&P 500 firms (but also shows that the relation is non-linear).

Major shareholders are often family members of the CEO for private firms (Hope et al., 2012a). There are interesting competing hypotheses when the CEO is related to the major shareholder. Because of the family relationship, these shareholders no longer act as independent monitors in disciplining CEOs' decisions. In addition, family-controlled firms are likely to suffer from greater horizontal agency costs. It may be easier for major shareholders, who are family members of the CEO, to extract private benefits from minority shareholders or other stakeholders. The reason it may be easier to extract these benefits is that major family owners typically have strong influence over choosing members of the board. Consequently, the monitoring effectiveness of the board may be impaired when its composition is determined primarily by the CEO's family. These arguments would support the idea that agency costs will increase when there is a family relation between the CEO and major shareholder (Hope et al., 2012a).

An alternative view is that family member CEOs are less likely to act in ways that opportunistically harm other family members. That is, installing a family member as the CEO could be a mechanism through which family-owned companies can increase their monitoring of management and reduce the need for external monitoring. If this effect dominates, the agency costs are smaller when the CEO is a family member because familial ties are likely to create closer alignment of the CEO's preferences with those of family owners.

In conclusion, vertical and horizontal agency costs supply opposite predictions for effects of family firms. In addition, there are strong differences in the degree to which families control business, to what extent the CEO comes from the dominant family, and in other institutional arrangements. In short, there is ample "tension" in terms of predictions and plenty of room for future research!

#### 3.1.2. Hope et al. (2012a) on agency conflicts in (private) family firms

Hope et al. (2012a) are interested in understanding how agency conflicts in private firms arise through ownership structures and family relationships. They analyze auditors' increase of effort and firms' choice of auditors in situations with higher level of agency conflicts. For a large sample of private Norwegian firms, they use data obtained through special permission by the government to measure direct and ultimate ownership for each shareholder as well as extended family relationships. Family relationships are measured based on marriage and blood lines, going back four generations and extending out to fourth cousin, and cover all shareholders, board members, and CEOs.

The authors find that (excess) audit fees, their proxy for audit effort in the face of agency conflicts, vary as hypothesized with firm-level characteristics related to ownership structures and family relationships. Specifically, they show that fees relate negatively to ownership concentration and to the extent of ownership by the second-largest shareholder. Audit fees also relate negatively to the portion of shares held by the CEO, consistent with ownership aligning the incentives of the CEO and other stakeholders. Audit fees are further positively associated with family relationships between the CEO and the major shareholder (a signal of reduced monitoring and a situation in which expropriation by the family/major shareholder is easier).

With respect to board independence, they find that audit fees decline as the number of board members related to the largest shareholder increases, consistent with fewer agency conflicts between owners and the board. In contrast, as the number of board members related to the CEO increases, fees increase, suggesting less board independence and greater agency conflicts.

Hope et al. (2012a) report two interesting sets of results for the demand for Big 4 auditor. First, for agency settings that are *not* CEO family-related, they observe results consistent with those obtained for the auditor effort tests. Specifically, the propensity to hire a Big 4 auditor increases as ownership concentration decreases, ownership of the second largest owner decreases, and the major shareholder's family influence on the board decreases. These results are consistent with the demand for a Big 4 auditor being greater in higher agency cost settings. They do not find significant evidence of a relation between hiring a Big 4 auditor and the fraction of shares owned by the CEO for the main tests but they do in sensitivity tests.

The authors find no association between the choice to hire a Big 4 auditor and CEO family-related agency variables. Specifically, there is no significant evidence that the demand for a Big 4 auditor is affected when a family relationship exists between the CEO and the major shareholder or as the number of board members related to the CEO increases. While some CEOs in family-related agency settings may wish to signal more credible reporting by hiring a Big 4 auditor, other CEOs in these settings may feel a Big 4 auditor is either

unnecessary given close family ties or unwanted because of the gains from extracting private benefits which could be reduced by a Big 4 audit.

#### 3.2. Institutional ownership

Institutional investors such as pension funds and mutual funds are often "large" shareholders. In addition, they are typically viewed as "sophisticated investors" in the literature. The extant theoretical literature generally predicts large institutional investors as an efficient form of corporate governance. However, large institutional holders are not using their own money to make investments. Thus, with regulatory constraints or lack of incentives, Coffee (1991) argues that institutional shareholders tend to be passive.

Prior research has documented that sophisticated investors behave differently from other, less informed investors (e.g., Callen et al., 2005). Sophisticated investors have superior abilities and consequently can learn better from experience (Bonner and Walker, 1994). Economic incentives are potentially important as well. Institutional investors have large investment portfolios and, therefore, have much more to gain or lose in absolute dollar terms from their investment decisions. Furthermore, the costs of engaging in in-depth firm analysis are lower for institutions, in part because of their superior access to databases and analytical tools.

Research documents the existence of distinct groups among institutions that differ in their objectives and information needs. Bushee (1998) classifies institutions into three groups – transient, dedicated, and quasi-indexers. "Transient" institutions have high portfolio turnover and highly diversified portfolio holdings. They focus on the short term and make investments based on the likelihood of short-term trading profits. According to Bushee (2001), the short investment horizons of transient investors create little incentive for them to gather information relevant to long-run value.

In contrast, "dedicated" investors and "quasi-indexers" focus on the long term and provide stable ownership to firms. Dedicated investors hold large stakes in a limited number of firms. Such ownership creates greater incentives to invest in monitoring management and to rely on information beyond current earnings to assess managers' performance. Quasi-indexers generally follow indexing and buy-and-hold strategies, and are characterized by high diversification. Although quasi-indexers follow a passive investment strategy, these investors may also have strong incentives to monitor management to ensure that it is acting in the best interest of the firm.

Many studies report results that are consistent with a superior ability of sophisticated investors to gather, analyze, and price information. Price (1998) finds that informed investors appear to make greater use of accounting disclosures and non-earnings information to form more precise earnings expectations. Bonner et al. (2003) document that sophisticated investors incorporate the information inherent in the relative accuracy of analyst forecasts to a greater extent than less informed investors. In addition, Bhattacharya et al. (2007) provide evidence that sophisticated investors demonstrate less behavioral bias in the way they process pro forma earnings information relative to more sophisticated investors. Finally, the efficiency of a firm's stock price is associated with the degree of sophistication of the firm's marginal investor (e.g., Bartov et al., 2000).

As an example of my own work that includes institutional investors, Chen et al. (2012) shows that the difference between closed-end country funds' net asset values and their trading prices (i.e., the fund discount) is positively associated with the earnings opacity of the underlying companies. In conditional analyses they further find that the positive relation between earnings opacity and fund discounts is weaker for those funds with a higher level of institutional ownership. In other words, investors who are better equipped at information acquisition than other investors are able to overcome some of the information disadvantage of being "non-local." In an earlier study, Callen et al. (2005) find that the variance contribution of foreign earnings increases with the level of investment by long-term (but not short-term) institutional investors.

To sum, there is strong evidence that institutional investors are an important class of large shareholders, in part because of their greater expertise in analyzing accounting information. There is also extensive evidence that there is important variation among the different classes of institutional investors. Thus, yet again we conclude that there is significant diversity among even subgroups of large shareholders.

# 3.3. State ownership<sup>7</sup>

#### 3.3.1. History/background

State ownership of enterprises is far from new and is not solely confined to Continental Europe or Asia. In practice most states have relied on the state to kick start growth or at least to protect fragile industries. More recently, Singapore is often viewed as starting the new kind of state capitalism. Lee Kuan Yew, its founding father, was a tireless advocate of "Asian values," by which he meant a mixture of family values and state control. However, state ownership is far from confined to quasi-authoritative states. In particular, many governments have found it desirable to have a tight control over their natural resources such as oil and gas. In China Deng Xiaoping transformed the economy by embracing globalization through creating special economic zones and inviting foreign companies in. He forced state enterprises to model themselves on Western companies and concentrated resources on national champions.<sup>8</sup>

#### 3.3.2. Scale and importance of state ownership

State ownership is prevalent around the world. The rich world still has a large number of state-owned or state-dominated companies. For example, France owns 85% of EDF, Japan owns 50% of Japan Tobacco, and Germany owns 32% of Deutsche Telekom. In total OECD state-owned enterprises have a combined value of almost \$2 trillion and employ 6 m people. However, state-owned enterprises are even more important in the emerging world. They make up most of the market capitalization of China's and Russia's stock markets and account for 28 of the emerging world's 100 biggest companies. Finally, in terms of industry focus, state ownership is especially noticeable in the energy sector, with the 13 biggest oil firms in the world all being state-backed (as is the world's biggest natural-gas company, Russia's Gazprom). These are also the companies in which governments tend to have the highest ownership stakes and the most direct control.

## 3.3.3. Types of state ownership and quality of management

Property rights theory argues that ownership and control rights should be given to the parties that make exante specific investments (e.g., Hart and Moore, 1996). According to the property rights theory SOE managers lack incentives to maximize corporate profitability, as the majority of the firm is owned by the state. Building on property rights theory, Hart et al. (1997) argue that privatized firms have a better incentive to minimize costs, but the systematic pursuit of profits may lead to poorer service quality. For example, following the privatization of railways in the UK and the Netherlands the quality of service visibly deteriorated. Schmidt (1996) argues that a trade-off exists for state-ownership. The benefit is that under state ownership the government has better information about the firm's management. The cost is that the government tends to interfere too much for political reasons.

In practice not all SOEs are created equal and there is evidence that governments are becoming more sophisticated owners. Only a handful of SOEs are still reporting directly to government ministries. <sup>10</sup> In contrast most governments prefer to exercise control through their ownership of shares. Sometimes they hold all the shares, but increasingly they prefer to dilute their shareholdings. <sup>11</sup> There is also evidence that SOEs have become more productive as a result of restructuring. For example, in China their return on assets increased from 0.7% in 1998 to 6.3% in 2006 (although accounting figures obviously can be manipulated – which should provide a promising area for future research). I discuss more China-related issues in Section 5.2.

<sup>&</sup>lt;sup>7</sup> The primary source for this subsection is The Economist Special Report "The Visible Hand" January 31, 2012.

<sup>&</sup>lt;sup>8</sup> Similarly, the post-Soviet disaster created a craving for order and the Russian government reasserted direct state control over "strategic" industries.

<sup>&</sup>lt;sup>9</sup> However, state firms can be found in almost any industry. China Mobile has 600 million customers. Saudi Basic Industries Corporation is a huge chemical company. Russia's Sberbank is Europe's third-largest bank. Dubai Ports is the world's third-largest ports operator. In addition to having ownership stakes in companies, governments are also important large owners through sovereign wealth funds.

<sup>&</sup>lt;sup>10</sup> Statoil of Norway is the world's 13th-biggest oil company by revenue. Norway also has the third-biggest sovereign-wealth fund, the Government Pension Fund. Both are required to behave like regular companies.

<sup>&</sup>lt;sup>11</sup> The United Nations Conference on Trade and Development defines a state-owned company as one in which the state owns more than 10% of the shares.

#### 3.4. Employee ownership

Compared with the other groups discussed in this section, employees are typically less significant as owners. However, there is also significantly less research on employee ownership and there are interesting cross-country variations in how firms are structured and hence in the importance of employees in the governance of the firms, both of which create opportunities for future studies. Although clearly a generalization, it is probably fair to state that employees, including labor unions, have a relatively stronger say in Continental Europe than elsewhere.

The pros and cons of employee ownership have inspired much debate in recent years (Bova et al., 2012). On the one hand, advocates of employee ownership cite evidence which suggests that employee ownership leads to increasing employee—manager goal alignment and productivity gains that are ultimately reflected in higher shareholder returns (Kruse, Blasi, and Park, 2009). On the other hand, contrasting empirical evidence suggests that giving non-manager employees too much ownership in the company can erode shareholder value (La Porta et al., 1997). 12

What is perhaps especially interesting about employees as an owner group is that there are three very different groups: managers, non-manager employees, and unions. There has been a fair amount of research on managerial ownership but much less on other employees and unions. Focusing on top managers' stock-based incentives, Nagar et al. (2003) find that stock-based incentives (and thus ownership) can reduce agency problems between managers and shareholders, and thus increase the incentives for managers to disclose information.

In contrast, Bova et al. (2012) investigate the role of non-manager employee ownership on voluntary disclosure. Specifically, they focus on the firm's employees as a group of stakeholders that have the potential to extract above-market rents from the firm and on employee ownership as a tool to mitigate this potential to extract rents. This provides for an interesting contrast between these two roles ("alignment" versus "rent extraction). The literature provides evidence that managers have an incentive to keep information asymmetric with the market if employees can extract above-market rents from the firm – for example, in cases where the employee base is highly unionized. The benefit to the strategy of disclosing less is that reduced transparency should weaken the employees' bargaining position. However, an opaque disclosure policy keeps information asymmetric with not only employees, but also investors and other stakeholders. Employee ownership can thus potentially play an important role in mitigating this tension. Cramton et al. (2008) provides analytical and empirical evidence that employee stock ownership leads to a greater propensity for employees to internalize the costs of labor disputes, which in turn reduces employees' incentive to extract rents through costly strikes, which are deadweight losses. The decrease in the incentive arises as employee compensation becomes more closely linked to the stock returns of the firm, leading to any costly negotiation frictions (e.g., extended negotiations or strikes) impacting employee compensation to a greater extent.

Bova et al. (2012) employ a number of proxies for voluntary disclosure and find that firms whose non-manager employees have strong bargaining power provide less voluntary disclosure whereas firms whose employees have larger equity stakes in the firm provide greater voluntary disclosure. Furthermore, the effect of employee ownership in generating better disclosure is particularly strong, the greater employees' negotiation leverage. In other words, employee ownership appears to benefit the firm by not only aligning goals between the firm and its employees, but by also increasing disclosure from the firm to *all* of its stakeholders by mitigating the firm's need to keep information opaque.

In conclusion, there is limited research on employee ownership and great potential for future research to take advantage of cross-country variation in such ownership.

<sup>&</sup>lt;sup>12</sup> For example, in 1995, United Airlines awarded employees 55% of the firm's equity in exchange for concessions on salaries and benefits. While the plan at the time was applauded by the US Federal government as an innovative way to heal the fractious nature of the union—management relationship, others remained skeptical of allowing a stakeholder that already contracted with the firm enough power to essentially control the firm's decision making (Bova et al., 2012).

#### 4. Private (i.e., unlisted) firms

#### 4.1. Importance of private firms

Private firms provide an important vehicle for economic growth around the world. More than 99% of limited liability companies, in most countries, are not listed on a stock exchange (e.g., Pacter, 2004; Chen et al., 2011). In the aggregate, non-listed firms have about four times more employees, three times higher revenues, and twice the amount of assets than do listed firms (Berzins et al., 2008). In 2008, *Forbes* reported that the 441 largest private companies in the United States accounted for \$1.8 trillion in revenues and employed 6.2 million people. Furthermore, according to the US Census Bureau, there are 29 million privately held companies in the United States, 7.6 million of which have paid employees, representing one-half of the nation's GDP (Hope et al., 2012b). Despite their obvious importance to the economy, there is limited extant research on private firms in general and in particular very little research related to accounting and auditing of such firms.

# 4.2. How are private firms different?

Private firms are different from publicly traded firms in several respects. Private firms are more closely held, have different governance, and have greater managerial ownership. Moreover, their major capital providers often have insider access to corporate information and typically take a more active role in management. With greater ownership concentration than in public corporations, large shareholders have a greater potential to take advantage of their controlling positions and direct private benefits for personal consumption, which is the typical problem of expropriation of minority shareholders and creditors (e.g., Morck et al., 1988).

# 4.3. The role of accounting and auditing in private firms

Some researchers take a strong position on whether accounting and auditing plays a lesser or stronger role in private compared with public firms. Personally I have no strong priors and I frankly believe there is not much evidence comparing the relative usefulness of accounting in these two sets of firms. In addition, I am not certain that it is fruitful to pursue such a line of inquiry per se.

The arguments in favor of reduced importance of accounting in private firms include the following. Most importantly, researchers often argue that there is lower demand for public accounting information in private firms as stakeholders may have access to private information. Furthermore, given the stronger ownership concentration, shareholder turnover is lower, and shareholders take a more active role in management, which some claim would reduce their reliance on financial statements for monitoring managers compared with public firms.

However, there are competing arguments also. Private firms typically have a weaker overall information environment compared with the relatively stronger disclosure environment of public firms. This suggests that, even if say the financial reporting quality (FRQ) is lower for private firms, accounting information could still play an important role since there are fewer competing sources of information. For example, McNichols and Stubben (2008) emphasize the role that accounting information plays in internal decision making. Small firms are unlikely to have management accounting systems that are separate from financial accounting, potentially enhancing the role of accounting in internal decision making (Chen et al., 2011). Finally, it is possible that the lack of analyst coverage and lower media coverage makes accounting information a relatively greater component of the overall information set used for decision making by insiders or outsiders (Chen et al., 2011).

With respect to the role of auditing, it is not obvious whether external auditors play a lesser or a stronger role in private firms than in public firms (Hope and Langli, 2010; Hope et al., 2011, 2012). On the one hand, Coffee (2005) discusses how the existence of controlling (i.e., especially large) shareholders can affect auditor independence. That is, Coffee (2005) argues that it is difficult for the auditor to escape the control of the party that she is expected to monitor. On the other hand, it is possible that the monitoring value of auditing is higher

<sup>&</sup>lt;sup>13</sup> For example, Indjejikian and Matejka (2009) highlight the importance of accounting information for private firms in compensation contracts.

in private firms because they are less vulnerable to takeovers and because they are required to disclose less accounting and non-accounting information than public firms (e.g., Lennox, 2005).

# 4.4. Some findings from the private firm setting 14

Not surprisingly, accounting research has focused primarily on properties of earnings and in particular on comparing financial reporting quality between private and public firms. There is clearly some tension in this question – whereas the "demand" perspective predicts higher FRQ for public firms, the "opportunism" perspective lead to the opposite prediction. While some research that has focused on specialized samples and industries have found higher FRQ in private firms, the large-sample investigations to date suggest that the demand hypothesis dominates and that FRQ is higher on average in public firms (e.g., Burgstahler et al., 2006 for European firms; Ball and Shivakumar, 2005 for UK firms; Hope et al., 2012b for US firms).

#### 4.4.1. Hope et al. (2012b)

There is very limited research to date on within-private firm variation in accounting. Hope et al. (2012b) provide the first exploration of cross-sectional variations in the FRQ of US private firms. They show that private firms with greater external financing needs and a greater presence of long-term debt have higher FRQ and greater conservatism.

More directly related to the topic of large shareholders, Hope et al. (2012) also investigate the expected impact of organizational form on FRQ from two perspectives – owner–manager separation and ownership dispersion. Because managers understand that their actions are not perfectly observable by the owner, managers have the ability to hide unfavorable performance by manipulating reported performance. Thus, firms which are more likely to suffer from agency costs (i.e., owner–manager separation) would be expected to have lower FRQ.

Owners are expected to take action by monitoring the activities of the manager. However, monitoring is costly, and owners are willing to incur monitoring costs only to the extent that the benefits outweigh the costs. These arguments suggest that when ownership dispersion is high, managers' activities are less likely to be closely monitored and therefore manipulation of reported performance is more likely to occur. Based on these arguments, firms with owner–manager separation and higher ownership dispersion should have lower FRQ.

However, Hope et al. (2012) also discuss competing arguments which suggest that separated ownership may positively affect private firms' FRQ. A controlling shareholder may have the ability to extract resources from the firm for personal consumption. These controlling shareholders may attempt to hide these activities from other stakeholders by manipulating reported performance. Such activities would lead to a positive relation between owner–manager separation and FRQ. In addition, the demand perspective would also predict a positive relation. In particular, when agency costs are higher, those contracting with the firm may demand more reliable financial information.

Hope et al.'s empirical findings indicate that private firms with more dispersed ownership (i.e., C corporations) have lower FRQ than other organizational structures as measured by three widely used FRQ proxies. These results are in line with the agency cost arguments (but not the demand arguments) described above.<sup>15</sup>

#### 4.4.2. Hope et al. (2011)

Hope et al. (2011) use a sample of private firms from 68 countries (mostly from emerging markets) obtained from the World Bank. They first show that firms with greater financial reporting credibility, operationalized as financial statements reviewed by an external auditor, experience significantly lower perceived problems in gaining access to external finance. More relevant to our topic, they additionally examine how this relation varies with ownership concentration and with cross-country institutional factors.

<sup>&</sup>lt;sup>14</sup> Recall that the findings of Hope et al. (2011) are summarized in Section 3.1.

<sup>&</sup>lt;sup>15</sup> They further find that C corporations exhibit higher conditional conservatism, which might be explained by the higher information asymmetry associated with C corporations creating the demand by investors, creditors, and others for more timely loss recognition. An alternative explanation for this finding relates to tax effects.

In their sample of private firms, the largest shareholder owns on average 74% of the shares and 69% of the firms have a controlling owner. Thus, their cross-country sample provides a rich setting for testing the effect of large shareholders.

Hope et al. (2011) are primarily interested in the effect of financial reporting credibility on financing constraints in the presence of a controlling owner. When there is a controlling shareholder, financial reporting credibility can play a greater role in reducing *costs* associated with agency and information problems. In other words, financial credibility matters more when there is a stronger need for it (i.e., high agency cost setting such as a controlling shareholder). In addition, there is no reason to expect increased financial credibility to reduce the *benefits* associated with a controlling owner. In fact, financial credibility may further improve monitoring and incentive alignment when a controlling owner exists. Therefore, regardless of whether the agency costs of a controlling owner outweigh the benefits, the authors unambiguously predict that financial credibility will have a greater effect on reducing financing constraints when a controlling owner exists. Their empirical results support this hypothesis – the effect of financial credibility on reducing perceived financing constraints is increasingly important in the presence of a controlling shareholder.

The study further examines the effect of cross-country variations in institutional factors. Recent cross-country literature places considerable emphasis on the adverse effects of private benefits of control (e.g., Dyck and Zingales, 2004). Countries which have better institutional properties (e.g., investor protection, legal enforcement, creditor rights, etc.) are better equipped to curb costs associated with private benefits of control. Hence, if agency costs related to private benefits of control are considered important by providers of external finance, then the mitigating role of financial credibility would likely be more pronounced in regimes with weaker institutions. In other words, financial credibility matters more when agency problems are more severe. Because private benefits of a controlling shareholder are less severe in countries with stronger institutions, there is less concern for these agency costs, and financial credibility is expected to have less of an effect. Consistent with these ideas, Hope et al. (2011) find that the impact of financial credibility in reducing financing constraints in the presence of a controlling owner is more pronounced in countries with weaker creditor rights.<sup>16</sup>

My conclusion from the limited extant research on private firms is that there is a wealth of opportunities for future research. For example, there are interesting data sources available for private firms in China, and such data bases could potentially be combined with data on political connections or other interesting issues relevant for the Chinese setting.

## 5. Country variations in the roles of large shareholders

#### 5.1. Introduction

Although it is primarily the more general agency cost arguments from Jensen and Meckling (1976) that have been cited by subsequent literature, it is also highly relevant for our discussion that Jensen and Meckling discuss the important *role which the legal system and the law play* in social organizations, especially, the organization of economic activity. In other words, we should not necessarily expect the same organizational structures or the same economic outcomes across different environments. As Jensen and Meckling (1976) explain, statutory laws sets bounds on the kinds of contracts into which individuals and organizations may enter without risking criminal prosecution. They focus on how the police and related powers of the state are used to enforce performance of contracts or to enforce the collection of damages for non-performance. The courts adjudicate conflicts between contracting parties and establish precedents. Such government activities affect both the kinds of contracts executed and the extent to which contracting is relied upon.

For a recent example of accounting research on how cross-country variations in the extent to which contracts are enforced matter to accounting outcomes, Dou et al. (2013) predict (based on incomplete contracting

<sup>&</sup>lt;sup>16</sup> Creditor rights are presumably one of the most important measures of legal protection associated with private firms.

theory) and find that firms that both reside in countries with weak contract enforceability and operate in industries with a greater need for relationship-specific investments tend to smooth reported earnings more. <sup>17,18</sup>

Perhaps even more interesting than how certain institutional factors vary across countries is what I will call "country peculiarities." These are differences across countries that are due to a multitude of factors that are difficult to summarize for researchers. Hence it may make sense to separately analyze certain issues related to large shareholders in specific countries. In the following I briefly mention some examples from China, India, and Japan/Korea.

#### 5.2. China: the role of state-owned enterprises

In China, the government plays a central role in corporate governance. In fact, the most obvious difference between China and many other countries is likely in the extent of state ownership of large companies. According to Li and Zhang (2010), China has by far the highest percentage of state-controlled firms in the countries they survey. Specifically, in their study the state is the "ultimate controller" for 63.15% of Chinese firms. In contrast, the corresponding figures for their other sample countries are: Singapore 23.5%, Germany 6.3%, France 5.11%, Hong Kong (included as a separate country in their study) 1.40%, Japan 0.80%, and the United Kingdom 0.08%. This huge difference in reliance on state ownership in China versus other countries suggests that the notion of large shareholders is likely quite different in China.

There are two types of state-owned listed companies in China: either the company is controlled by a parent (holding) company or the majority shares of the listed company are held by a state asset management (operational) company (Tomasic and Andrews, 2007). Tian (2001) finds that the government is the majority shareholder of 31.4% of the Chinese public listed companies. Thus, the state is often the largest shareholder for publicly traded companies resulting in many state-owned enterprises in the market. However, contradicting theoretical and empirical evidence exist on whether the state ownership is beneficial in creating market value.

Property rights theory argues that ownership and control rights should be given to the parties that make exante specific investments (e.g., Hart and Moore, 1996). According to the property rights theory SOE managers lack incentives to maximize corporate profitability, as the majority of the firm is owned by the state. However, given the unique government structure in China, it is difficult to draw any solid predictions through just examining the theoretical literature.

Official statistics suggest that about one third of Chinese SOEs are loss makers, another third either break even or making losses and the remaining one third are marginally profitable (Bai, Liu, Lu, Song, and Zhang, 2003). Bai, Liu, Lu, Song, and Zhang (2003) empirically show that a firm's performance is negatively affected when the largest shareholder is the government. Tomasic and Andrews (2007) conducted interviews with various corporate participants and outline how there is a lack of minority shareholder protection in the presence of state ownership in China.

However, state ownership can also be beneficiary. For example, state ownership can provide long-term and stable ownership and ensure financing is available also during crisis periods. Both Tian (2001) and Hess et al. (2010) conclude that there is a U-shaped relation between government shareholding and market value. The detected U-shape implies that firms dominated by the state players continue to maintain a greater respect by the market and outperform those with lower levels of state ownership. However, the effects of state ownership in mitigating minority shareholder expropriation or manipulation of the market at lower levels of state ownership are limited.

Whether state ownership is "good or bad" is not as relevant to accounting researchers as how it can affect interesting outcomes. In the following I very briefly review some recent studies that focus on ownership issues (and state ownership in particular) in China.

You and Du (2012) employ both agency and resource dependency theory to predict involuntary CEO dismissal and subsequent firm performance in Chinese firms. They find that board monitoring mechanisms

<sup>&</sup>lt;sup>17</sup> Dou et al. (2012) further decompose income smoothing into "garbled" and "informational" components and find that results are driven by the informational component of income smoothing.

<sup>&</sup>lt;sup>18</sup> Although not the focus of this article, I would recommend authors to look beyond the standard La Porta et al. measures when looking for country-level variations. This is not a criticism of the La Porta et al. measures; however there are many other interesting variables and new websites available.

explain very little of the outcomes, but political ties with government officials at the state, provincial, county, and city levels are highly predictive of CEO turnover and ultimately firm performance. In other words, they conclude that political ties overrule economic norms in China, lending stronger support for the resource dependency perspective and challenging the agency cost perspective in the Chinese context.

Li and Zhang (2010) use Shanghai National Accounting Institute's Chinese Firms' social responsibility ranking and observe a negative relation between corporate ownership dispersion and corporate social responsibility for a state-owned firm; whereas a positive relation exists for non-state-owned firms. The authors attribute their finding to the large degree of political interference in state-owned firms.<sup>19</sup>

Pi and Lowe (2011) find no association between Chinese CEO turnover and the percentage of shares held by CEOs. They interpret the finding to mean that CEOs do not derive significant power from their shareholder status in China. Furthermore, they find that CEOs in state-owned firms are significantly less likely to be replaced involuntarily and conclude that CEOs in state-owned firms are likely enjoying more discretion because state-owned firms have weaker corporate governance mechanisms and strong political connections.<sup>20</sup>

#### 5.3. India: the role of business groups

Transaction cost theory suggests that the optimal structure of a firm depends on its institutional context. Khanna and Palepu (2000) discuss how diversified business groups dominate private sector activity in many emerging markets and in particular in India.<sup>21</sup> The typical Indian business groups are collections of publicly traded firms in a wide variety of industries, with a significant amount of common ownership and control, usually by a family. Prior US literature has documented that businesses affiliated with diversified firms underperform their focused competitors. Among the reasons cited for the underperformance of diversified corporations are inappropriate allocation of decision rights, inefficient allocation of capital, and poor internal governance.

In countries such as India there are a variety of market failures, caused by information and agency problems. For example, firms often provide limited financial disclosure and often have weak corporate governance and control. In addition, intermediaries such as financial analysts or the financial press are not fully evolved and securities regulations and related enforcement are weaker than in Western countries. There is thus a potential for diversified business groups that can act as an intermediary between individual entrepreneurs and imperfect markets.

Khanna and Palepu (2000) analyze the performance of affiliates of diversified Indian business groups relative to unaffiliated firms. They find that accounting and stock market measures of firm performance initially decline with group diversification and subsequently increase once group diversification exceeds a certain level. Interestingly, unlike US conglomerates' lines of business, affiliates of the most diversified business groups outperform unaffiliated firms. Of the potential sources of performance effects of group affiliation, Khanna and Palepu (2000) find the strongest effect related to group-affiliated firms' access to international capital markets (presumably due to the track record of the group as a whole).

# 5.4. Japan and Korea: the role of keiretsus and chaebols

The ownership structure of Japanese (and also many Korean) firms is typically highly concentrated among corporate stockholders with financial institutions occupying a majority of the stock holdings (e.g., Douthett

<sup>&</sup>lt;sup>19</sup> Bai et al. (2004) use a panel date of 32 two-digit industries in 29 Chinese regions over the period of 1985–1997 and find that the degree of regional specialization is low for firms with larger shares of state ownership. The finding indicates that local governments have strong incentives to protect their industries.

<sup>&</sup>lt;sup>20</sup> Chang and Wong (2004) examine local party committees' role in China's economic reform and suggest publicly listed Chinese firms can improve firm performance through decreasing local political party members' existing control level, supporting the grabbing hand theory. In addition, Yang et al. (2011) also support the idea of decreasing state-ownership in China to improve corporate governance. Huang and Xu (2009) study large blocks of share transfer in China and find a positive correlation between private benefit of control and block price but a negative relation between trading restrictions and block price. Moreover, private institutions offer a higher price than state-owned institutions.

<sup>&</sup>lt;sup>21</sup> Almeida and Wolfenzon (2006) provide a theoretical analysis of family business groups and in particular provide a new rationale for pyramidal ownership. A pyramid allows a family to access all retained earnings of a firm it already controls to set up a new firm, and to share the new firm's no diverted payoff with share-holders of the original firm.

<sup>&</sup>lt;sup>22</sup> Rather surprisingly, there is very limited accounting research using Indian data, which suggests opportunities for future work.

and Jung, 2001). A significant portion of Japanese industrial firms' ownership is represented by small groups of enterprises – *keiretsus* – composed of firms in different industries. These firms are interrelated through cross holdings of equity ownership and generally rely on a large commercial bank for their primary banking needs. The keiretsu firms maintain close financial and personal ties through cross-shareholding, credit holding, interlocking corporate directorates within the group, and a variety of business transactions.

On the one hand, the *keiretsu* relationship has the potential to increase the monitoring of managerial performance. On the other hand, the keiretsu relationship may decrease the effectiveness of monitoring. Managers may entrench in an inefficient, low-effort arrangement in which managers protect each other in the market for corporate control, resulting in an anti-competitive and exclusionary environment.

As but one example from Japan, Douthett and Jung's (2001) find that Japanese keiretsu firms have higher earnings response coefficients (their proxy for informativeness of earnings) than those of non-keiretsu firms. In addition, the ERC increases as the strength of the keiretsu relationship increases. Finally, discretionary accruals by keiretsu firms are smaller than discretionary accruals of non-keiretsu firms. Douthett and Jung (2001) conclude that the monitoring ability of the keiretsu improves the informativeness of earnings.

Business practices in Korea are similar although not identical to Japan. There is widespread use of pyramid ownership structures and cross-holdings among firms that belong to a business group. This type of corporate structure allows controlling shareholders to exercise full control over a firm despite holding a relatively small portion of its cash flow rights (e.g., Baek et al., 2006). Such a divergence between ownership and control raises concerns about the degree to which the controlling shareholders siphon resources out of firms to increase their wealth, that is, the degree to which the controlling shareholders engage in tunneling.

Baek et al. (2006) find that Korean chaebol issuers involved in intragroup deals set the private securities offering prices to benefit their controlling shareholders. They also find that chaebol issuers realize an 8.8% higher announcement return than do other types of issuers if they sell private securities at a premium to other member firms, and if the controlling shareholders receive positive net gains from equity ownership in issuers and acquirers. These results are consistent with tunneling within business groups.<sup>23</sup>

The overall conclusion from this section is that standard agency cost theory predictions do not necessarily apply to all countries around the world. The legal, cultural, and other contexts in which ownership operates will likely influence the governance impact of large shareholders. It is incumbent upon us as researchers to both understand the environment that we are studying and to make use of such variations in the environment to come up with interesting new research questions to pursue. The good news is that there should be plenty of opportunities for future research!

#### 6. Brief concluding remarks

This article has focused on highlighting the heterogeneity of large shareholders. Specifically, I discuss the importance of families, institutions, governments, and employees as shareholders. The roles of each of these is likely different and thus it is prudent to go beyond overly general notions of "large shareholders" and instead consider exactly who the owners are, and also suggests that further research on specific shareholder type effects on accounting outcomes would be welcome. I also emphasize that if researchers are really interested in researching ownership concentration, private firms may be the most fertile ground, both because these firms tend to have more concentrated ownership and because there is considerably less prior research on these economically important firms compared with publicly traded companies.

But perhaps most importantly, I would encourage readers to use their imagination and not just "follow the bandwagon" in terms of choosing research topics. I would recommend reading outside of accounting (both to find interesting topics and to bring methodological advances into accounting) and to follow closely what is happening in practice. There are many exciting research opportunities in China and I very much look forward to future issues of *CJAR* and to attending future conferences in China.

<sup>&</sup>lt;sup>23</sup> Bae et al. (2008) examine intragroup propping within Korean chaebols. They find that the announcement of increased (decreased) earnings by a chaebol-affiliated firm has a positive (negative) effect on the market value of other non-announcing affiliates. This finding is consistent with the market's *ex ante* valuation of intragroup.

#### References

- Adams, R.B., Hermalin, B.E., Weisbach, M.S., 2010. The role of boards of directors in corporate governance. A conceptual framework and survey. Journal of Economic Literature 48, 58–107.
- Ali, A., Chen, T.-Y., Radhakrishnan, S., 2007. Corporate disclosures by family firms. Journal of Accounting and Economics 44 (1–2), 238–286
- Almeida, H.V., Wolfenzon, D., 2006. A theory of pyramidal ownership and family business groups. Journal of Finance 61 (2637–2680),
- Anderson, R.C., Reeb, D.M., 2003. Founding-family ownership and firm performance. Evidence from the S&P 500. The Journal of Finance 58 (3), 1301–1327.
- Bae, G.S., Cheon, Y.S., Kang, J.-K., 2008. Intra-group propping: evidence from the stock-price effects of earnings announcement by Korean business groups. Review of Financial Studies 21, 2015–2060.
- Baek, J.-S., Kang, J.-K., Lee, I., 2006. Business groups and tunneling: evidence from private securities offerings by Korean chaebols. Journal of Finance 61, 2415–2449.
- Bai, C.E., Liu, Q., Lu, J., Song, F.M., Zhang, J.X., 2003. Corporate governance and market valuation in China. Working Paper, University of Hong Kong.
- Bai, C.E., Du, Y., Tao, Z., Tong, S.Y., 2004. Local protectionism and regional specialization: evidence from China's industries. Journal of International Economics 63, 397–417.
- Ball, R., Shivakumar, L., 2005. Earnings quality in UK private firms: comparative loss recognition timeliness. Journal of Accounting and Economics 39 (1), 83–128.
- Bartov, E., Radakrishnan, S., Krinsky, I., 2000. Investor sophistication and patterns in stock returns after earnings announcements. Accounting Review 75 (1), 43–63.
- Bertrand, M., Schoar, A., 2006. Role of family in family firms. Journal of Economic Perspectives 20 (2), 73-96.
- Berzins, J., Bøhren, Ø., Rydland, P., 2008. Corporate Finance and Governance in Firms with Limited Liability: Basic Characteristics. Research Report No. 1, Centre for Corporate Governance Research, Norwegian School of Management.
- Bhattacharya, N., Black, E.L., Christensen, T.E., Mergenthaler, R.D., 2007. Who trades on pro forma earnings information? The Accounting Review 82 (3), 581–619.
- Biddle, G., Hilary, G., 2006. Accounting quality and firm-level capital investment. The Accounting Review 81, 963-982.
- Bloom, N., Van Reenen, J., 2007. Measuring and explaining management practices across firms and countries. The Quarterly Journal of Economics 122 (4), 1351–1408.
- Bonner, S.E., Walker, P., 1994. The effects of instruction and experience on the acquisition of auditing knowledge. Accounting Review 69 (1), 157–178.
- Bonner, S.E., Walther, B.R., Young, S.M., 2003. Sophisticated and unsophisticated investors' reactions to analysts' forecast revisions conditional on factors that are associated with forecast accuracy. Accounting Review 78 (3), 679–706.
- Bova, F., Dou, Y., Hope, O.-K., 2012. Employee Ownership and Firm Disclosure. Working Paper, University of Toronto.
- Burgstahler, D., Hail, L., Leuz, C., 2006. The importance of reporting incentives: earnings management in European private and public firms. The Accounting Review 81, 983–1016.
- Bushee, B., 1998. The influence of institutional investors on myopic R&D investment behavior. Accounting Review 73 (3), 305-333.
- Bushee, B., 2001. Do institutional investors prefer near-term earnings over long-run value? Contemporary Accounting Research 18, 207–246.
- Callen, J., Hope, O.-K., Segal, D., 2005. Domestic and foreign earnings, stock return variability, and the impact of investor sophistication. Journal of Accounting Research 43, 377–412.
- Chang, E., Wong, S., 2004. Political control and performance in China's listed firms. Journal of Comparative Economics 32, 617–636.
- Chen, S., Chen, X., Cheng, Q., 2008. Do family firms provide more or less voluntary disclosure? Journal of Accounting Research 46 (3), 499–536.
- Chen, F., Hope, O.-K., Li, Q., Wang, X., 2011. Financial reporting quality and investment efficiency of private firms in emerging markets. The Accounting Review 86 (4), 1255–1288.
- Chen, F., Hope, O.-K., Li, Q., Wang, X., 2012. Earnings Opacity and Closed-End Country Fund Discounts. Working Paper, University of Toronto, Wuhan University, and University of Hong Kong.
- Claessens, Stijn, Djankov, Simeon, Fan, Joseph P.H., Lang, Larry H.P., 2002. Disentangling the incentive and entrenchment effects of large shareholdings. Journal of Finance 57 (6), 2741–2771.
- Coffee Jr., J.C., 1991. Liquidity versus control: the institutional investor as corporate monitor. Columbia Law Review 91, 1277-1368.
- Coffee Jr., J.C., 2005. A theory of corporate scandals: why the USA and Europe differ. Oxford Review of Economic Policy 21 (2), 198–211. Cramton, P., Mehran, H., Tracy, J., 2008. ESOP Fables: The Impact of Employee Stock Ownership Plans on Labor Disputes. FRB of
- Cramton, P., Mehran, H., Tracy, J., 2008. ESOP Fables: The Impact of Employee Stock Ownership Plans on Labor Disputes. FRB of New York Staff Report No. 347.
- Cronqvist, H., Fahlenbrach, R., 2009. Large shareholders and corporate policies. Review of Financial Studies 22 (10), 3941–3976.
- Cronqvist, H., Nilsson, M., 2003. Agency costs of controlling minority shareholder. Journal of Financial and Quantitative Analysis 38 (4), 695–719.
- Dou, Y., Hope, O.-K., Zou, Y., 2012. Large Shareholders and Accounting Choices. Working Paper, University of Toronto and New York University.
- Dou, Y., Hope, O.-K., Thomas, W.B., 2013. Relationship-specificity, contract enforceability, and income smoothing. The Accounting Review, forthcoming.

Douthett Jr., E.B., Jung, K., 2001. Japanese corporate groupings (keiretsu) and the informativeness of earnings. Journal of International Financial Management and Accounting 12 (2), 133–159.

Dyck, A., Zingales, L., 2004. Private benefits of control: an international comparison. Journal of Finance 59, 537-600.

Ellul, A., Guntay, L., Lel, U., 2009. Blockholders, Debt Agency Costs and Legal Protection. Working Paper, FRB International Finance Discussion Paper No. 908.

Fama, E., 1980. Agency problems and the theory of the firm. Journal of Political Economy 88 (2), 288-307.

Fama, E., Jensen, M., 1983. Separation of ownership and control. Journal of Law and Economics 26 (2), 327-349.

Gjesdal, F., 1981. Accounting for stewardship. Journal of Accounting Research 19 (1), 208-231.

Gogineni, S., Linn, S.C., Yadav, P.K., 2010. Empirical Evidence on Ownership Structure, Management Control and Agency Costs. Working Paper, University of Oklahoma.

Hart, O., Moore, J., 1996. The governance of exchanges: members' cooperatives versus outside ownership. Oxford Review of Economic Policy 12 (4), 53–69.

Hart, J., Shleifer, A., Vishny, R.W., 1997. The proper scope of government: theory and an application to prisons. Quarterly Journal of Economics 112 (4), 1126–1161.

Hess, K., Gunasekarage, A., Hovey, M., 2010. State-dominant and non-state-dominant ownership concentration and firm performance. Evidence from China. International Journal of Managerial Finance 6 (4), 264–289.

Hope, O.-K., Thomas, W.B., 2008. Managerial empire building and firm disclosure. Journal of Accounting Research 46, 591-626.

Hope, O.-K., Langli, J.C., 2010. Auditor independence in a private firm and low litigation setting. The Accounting Review 85 (2), 573–605. Hope, O.-K., Thomas, W.B., Vyas, D., 2011. Financial credibility, ownership, and financing constraints in private firms. Journal of International Business Studies 42 (7), 935–957.

Hope, O.-K., Langli, J.C., Thomas, W.B., 2012a. Agency conflicts and auditing in private firms. Accounting, Organizations, and Society 37 (7), 500–517.

Hope, O.-K., Thomas, W.B., Vyas, D., 2012. Financial Reporting Quality in US Private and Public Firms. Working Paper, University of Toronto, University of Oklahoma, and University of Minnesota.

Huang, Z., Xu, X., 2009. Marketability, control, and the pricing of block shares. Journal of Banking and Finance 33, 88-97.

Indjejikian, R., Matejka, M., 2009. CFO fiduciary responsibilities and annual bonus incentives. Journal of Accounting Research 47 (4), 1061–1093

Jensen, M.C., Meckling, W.H., 1976. Theory of the firm: managerial behavior, agency costs and ownership structure. Journal of Financial Economics 3 (4), 305–360.

Khanna, T., Palepu, K., 2000. Is group affiliation profitable in emerging markets? An analysis of diversified Indian business groups. Journal of Finance 55 (2), 867–891.

Kruse, P., Blasi, J., Park, R., 2009. Shared Capitalism in the US Economy: Prevalence, Characteristics, and Employee Views of Financial Participation in Enterprises. Shared Capitalism at Work: Employee Ownership, Profit and Gain Sharing, and Broad-based Stock Options, University of Chicago Press (Chapter 1).

La Porta, R.L., Lopez-De-Silanes, F., Shleifer, A., 1999. Corporate ownership around the world. The Journal of Finance 54 (2), 471–517. Lennox, C.S., 2005. Management ownership and audit firm size. Contemporary Accounting Research 22 (1), 205–227.

Li, W., Zhang, R., 2010. Corporate social responsibility, ownership structure, and political interference. Evidence from China. Journal of Business Ethics 96, 631–645.

McNichols, M.F., Stubben, S.R., 2008. Does earnings management affect firms' investment decisions? The Accounting Review 83, 1571–1603.

Morck, R., Shleifer, A., Vishny, R.W., 1988. Management ownership and market valuation: an empirical analysis. Journal of Financial Economics 20, 293–315.

Morck, R.K., Stangeland, D.A., Yeung, B., 2000. Inherited Wealth, Corporate Control, and Economic Growth? In: Morck, Randall K. (Ed.), Concentrated Corporate Ownership, NBER Conference Volume. University of Chicago Press, Chicago.

Nagar, V., Nanda, D., Wysocki, P., 2003. Discretionary disclosures and stock-based incentives. Journal of Accounting and Economics 34, 283–309.

Nagar, V., Petroni, K., Wolfenzon, D., 2011. Governance problems in close corporations. Journal of Financial and Quantitative Analysis 46 (4), 943–966.

Pacter, P., 2004. Financial reporting: SMEs-will the GAAP widen for SMEs? Accountancy 133, 118-119.

Pagano, M., Roell, A., 1998. The choice of stock ownership structure; agency costs, monitoring and the decision to go public. Quarterly Journal of Economics 113, 187–225.

Pi, L., Lowe, J., 2011. Can a powerful CEO avoid involuntary replacement? – An empirical study from China. Asia Pacific Journal Management 28, 775–805.

Price, R.E., 1998. Price Responsiveness of Informed Investors to Increases in Financial Statement Disclosure Quality. Working Paper, College of William and Mary.

Schmidt, K., 1996. The costs and benefits of privatization: an incomplete contracts approach. Journal of Law, Economics, and Organization 12 (1), 1–24.

Shleifer, A., Vishny, R.W., 1986. Large shareholders and corporate control. Journal of Political Economy 94 (3), 461-488.

Shleifer, A., Vishny, R.W., 1997. A survey of corporate governance. Journal of Finance 52 (2), 737–783.

Smith, A., 1776 (reprinted 1937), The wealth of nations, Cannan edition. Modern Library, New York.

Sraer, D., Thesmar, D., 2007. Performance and behavior of family firms: evidence from the French stock market. Journal of the European Economic Association 5 (4), 709–751.

Tian, L., 2001. Government Shareholding and the Value of China's Modern Firms. Working Paper, London Business School. Tomasic, R., Andrews, N., 2007. Minority Shareholder Protection in China's top 100 Listed Companies. Working Paper, Australian Research Council.

Wang, D., 2006. Founding family ownership and earnings quality. Journal of Accounting Research 44 (3), 619-656.

Yang, J., Chi, J., Young, M., 2011. A review of corporate governance in China. Asian Pacific Economic Literature.

You, J., Du, G., 2012. Are political connections a blessing or a curse? Evidence from CEO turnover in china. Corporate Governance: An International Review 20 (2), 179–194.